

Towards Technology-News-Driven Business Cycles

Paola Di Casola and Spyridon Sichlimiris

November 2018

WORKING PAPERS ARE OBTAINABLE FROM

www.riksbank.se/en/research

Sveriges Riksbank • SE-103 37 Stockholm Fax international: +46 8 21 05 31 Telephone international: +46 8 787 00 00

The Working Paper series presents reports on matters in the sphere of activities of the Riksbank that are considered to be of interest to a wider public. The papers are to be regarded as reports on ongoing studies and the authors will be pleased to receive comments.

The opinions expressed in this article are the sole responsibility of the author(s) and should not be interpreted as reflecting the views of Sveriges Riksbank.

Towards Technology-News-Driven Business Cycles^{*}

Paola Di Casola Sveriges Riksbank Spyridon Sichlimiris Örebro University & Sveriges Riksbank[†]

Sveriges Riksbank Working Paper Series No. 360 November 2018

Abstract

We identify an inflationary technology news shock as the leading source of business cycle variations for the postwar U.S. economy. This shock acts like a demand shock: it induces strong positive comovement in real quantities - GDP, consumption, investment - and weak positive comovement between real quantities and inflation, contrary to the view that anticipated technological innovations reduce inflation. The technology news shock became the predominant source of the business cycle from the 80's. The reason is that anticipated improvements in future technology lead to improvements in financing conditions. The monetary policy response to these anticipations is contractionary in the short run, independently of the sample period. However, the response is more short-lived from the 80's than before and with respect to other non-technology shocks. Finally, the inclusion of sentiment, uncertainty and TFP measurement error shocks does not affect the importance of the technology news shock.

JEL Codes: E20, E31, E32, E44, E52, O33

Keywords: total factor productivity, business cycle, technology news shocks, demand shocks, financial sector transmission, monetary policy.

*We are grateful to Jesper Lindé and Karl Walentin for extremely valuable comments. In addition, we thank Tobias Broer, Daniel Buncic, Tore Ellingsen, Daria Finocchiaro, Paolo Giordani, André Kurmann, Jean-Paul L'Huillier, Lars Ljungqvist, Conny Olovsson and seminar participants at Sveriges Riksbank and Greater Stockholm Macro Group for useful suggestions and discussions. Spyridon Sichlimiris thanks the Jan Wallander and Tom Hedelius Foundation for financial support. The opinions expressed in this article are the sole responsibility of the authors and should not be interpreted as reflecting the views of Sveriges Riksbank. All remaining errors are our own.

[†]Corresponding author: Spyridon.Sichlimiris@oru.se

1 Introduction

What drives business cycle fluctuations? The modern analysis of business cycles is centered around the study of "shocks". Shocks are considered to be exogenous forces that cause business cycle variation and are distinguished based on their origination (technology, policy, financial, etc...) and whether they are anticipated or unanticipated.

During the last five decades, the study of economic fluctuations has focused mostly on unanticipated shocks as the main cause of these variations. The idea that business cycles may be driven not only by surprises but also by shocks to expectations about future fundamentals has a long history. However, only during the last two decades these ideas have resurged. The seminal empirical work by Beaudry and Portier (2006) assigns a leading role to the technology (TFP) news shock as the driver of the business cycle, because it can generate strong positive correlation among real macroeconomic quantities and hours worked. Since then, research on news-driven business cycles has been not only empirical, but also model-based.¹ Using estimated DSGE models it is difficult to generate significant technology news-driven business cycles (Fujiwara et al. (2011), Khan and Tsoukalas (2012), Schmitt-Grohé and Uribe (2012)). In addition, other empirical studies (Barsky and Sims (2011), Forni et al. (2014), Kurmann and Sims (2017)) find that the technology news shock cannot be the main source of economic fluctuations, since it does not generate comovement among macroeconomic quantities and hours worked.

In this paper, we show that the major U.S. business cycle shock for the post World War II period is an anticipated technology shock. Differently from the aforementioned literature, the technology news shock we identify behaves like a demand shock. It induces strong positive comovement among real macroeconomic quantities - output, consumption, investment - and weak positive comovement among these real quantities and inflation. This shock established its prominence from the 80's. Before the 80's, based on the forecast error variance decomposition, it was as important as other technology or non-technology shocks. We argue that the technology news shock is more important for the business cycle from the 80's because its transmission through the financial sector is strengthened. A positive technology news shock eases financing conditions only from the 80's, as found also by Gambetti et al. (2018). In particular, the Baa corporate spread falls after a positive technology news shock from the 80's, and does not react before the 80's. At the same time, the response of monetary policy in the short run is found to be contractionary, independently of the period of study. However, the response of the real rate is more short-lived from the 80's compared to the period before the 80's and to the non-TFP shocks.

¹For an extensive literature review, see Lorenzoni (2011) and Beaudry and Portier (2014).

The period before the 80's in the U.S. was characterized by strict financial regulation. In 1980 the Depository Institutions Deregulation and Monetary Control Act was signed. This act gradually phased out restrictions for banks, while also improving the Federal Reserve's control of monetary policy, by extending reserve requirements to depository institutions.² These changes in the monetary and financial regulatory environment may have played a role in the improved transmission of the technology news shock to the real economy.

Methodologically, the current paper extends, modifies and enriches the structural vector error correction model initially studied by Beaudry and Portier (2006). To be more specific, we consider a core vector error correction model with four variables: TFP, consumption, output, investment.³ With this core system we identify four shocks based on zero short-run and long-run restrictions. The first two shocks are technology shocks, the surprise technology shock and the technology news shock, and the other two shocks are non-technology shocks, one of which is allowed to affect permanently all real macroeconomic quantities. The identification of the surprise technology shock is based on long-run restrictions á la Blanchard and Quah (1989). The technology news shock is assumed to have no impact effect on TFP but is allowed to affect it in the long run.

Compared to Beaudry and Portier (2006), we identify the technology news shock through consumption instead of stock prices. The reasons for this choice are the following. First, the use of stock prices is problematic due to the statistical properties of the series, that change around the 80's.⁴ This is not the case for the series of consumption. Second, as explained by Ramey (2016), current stock prices do not necessarily react positively to news about future TFP improvements. This is the case when major technological innovations make the existing technology less valuable, hence may have a negative effect on stock prices. Moreover, innovations that take place in small private companies take time to be reflected in the stock market index S&P 500, that represents large companies. Hence, we choose to identify "news" through consumption. Cochrane (1994) was the first to do so and Beaudry and Portier (2014) have done it as robustness. Indeed, consumption contains information about future fundamentals, because it aggregates the behaviour of forward-looking consumers.⁵

²For more information on the act and its impact on the Federal Reserve, see McNeill and Rechter (1980) and https://www.federalreservehistory.org/essays/monetary_control_act_of_1980.

³We show that our core system does not suffer from serious non-fundamentalness problems.

⁴As discussed in Di Casola and Sichlimiris (2018), the series of real per-capita stock prices behaves like a unit root without drift up to the 80's and like a unit root with drift from the 80's. Using stock prices in a VEC model requires to account for a structural break and makes the identification of the news shock sensitive to the data period used for the analysis.

⁵"Ask a consumer about next year's GDP and he will answer 'I don't know.' But he may know that his factory is closing, and hence he is consuming less. This idiosyncratic shock is correlated with future GDP. Summing over consumers, aggregate consumption can reveal information about future aggregate activity, although neither consumers in the economy nor economists who study it can name what the crucial pieces of information are" (Cochrane (1994), pag. 350).

Through the core system we establish that the technology news shock is the main driver of the business cycle, since it accounts for two thirds of the variation of output. As mentioned above, this shock became the dominant driving force of the business cycle from the 80's. In order to highlight the changes in the propagation of this shock, we extend our core model. We do so by including variables connected to prices (e.g CPI inflation), interest rates (e.g. real Treasury bill) and financial sector conditions (e.g. corporate Baa spread). Based on these extensions, we find that the technology news shock behaves like a demand shock.

Our empirical study is the first one to obtain the positive response of inflation after a technology news shock. The empirical literature on news shocks (Barsky and Sims (2011), Forni et al. (2014), Kurmann and Sims (2017) and Fève and Guay (2018)) obtains a negative response. The negative response is at odds with standard New Keynesian models and has motivated research that adds features into the canonical model to reproduce the empirical finding. Most notable work in this direction comes from Jinnai (2013), Kurmann and Otrok (2014), Barsky et al. (2015). The fact that our identified technology news shock is inflationary means that the empirical evidence need not necessarily be at odds with standard New Keynesian models.

Kurmann and Sims (2017) and Bouakez and Kemoe (2017) have highlighted the potential problems coming from measurement errors in utilization-adjusted TFP. We further investigate whether measurement problems can produce an upward bias in the importance of the news shock, by introducing a TFP measurement error shock. We find that our results carry on to this modified system. In addition, our results regarding the dynamics of the news shock are robust to introducing confidence and uncertainty shocks.

Finally, our results show that the second most important shock for the business cycle is a non-technology, non-inflationary permanent shock, that accounts for 25-30% of the output variation. Given that this shock is assumed not to affect technology, but it has a permanent effect on output and hours worked, one candidate explanation is that it represents a labour supply shock, as first described in Shapiro and Watson (1988). Indeed, we find support for this interpretation for two main reasons. First, it is a significant source of variation of hours worked in the long run. Second, it does not behave like a demand shock. We conclude that business cycle fluctuations can have a parsimonious two-shock representation: the technology news shock and the labour supply shock. The latter one becomes less important for the business cycle over time. Therefore, we are heading towards technology-news driven business cycles.

Related literature and comparison. As we have already mentioned, there are various shocks that have been studied as possible causes of the business cycle. It is beyond our scope to summarize this research. We refer the reader to recent and past attempts to

summarize these developments (Cochrane (1994), Ramey (2016)). Due to the fact that the technology news shock is responsible for two thirds of the variation in output, we will focus most of our discussion on papers that identify technology news shocks and may or may not identify other shocks.

We date the beginning of the modern literature on technology news shocks to Beaudry and Portier (2006). Afterwards, one strand of the literature has tried to gauge the importance of news shocks with estimated DSGE models and another strand with VAR-based approaches. The distinction would not be important, if these two different approaches were not producing contradictory findings. To be more specific, the VAR-based approach to news shocks led more support to the view that technology news shocks matter for the business cycle. In contrast, the DSGE-based approach cannot find support for the the technology news shock. Fujiwara et al. (2011) and Khan and Tsoukalas (2012) use a New Keynesian model, while Schmitt-Grohé and Uribe (2012) rely on a real business cycle model. They find that news shocks are important for the business cycle, but not the ones about technology.⁶ There is one recent notable exception, Görtz and Tsoukalas (2017). They find that technology news shocks matter for the business cycle based on a New Keynesian model with financial frictions, estimated on post-1990 data. The crucial difference with other estimated DSGE models is that the transmission of the technology news shock is amplified through a financial intermediation channel. This line of thought can be consistent with our empirical evidence. We have argued that the financial propagation of the technology news shock is key for the technology news-driven business cycle. Thus, we are heading towards a consensus between evidence coming from (cointegrated) VAR systems and estimated DSGE models.

In terms of VAR models, two main approaches have emerged. The first one, pioneered by Beaudry and Portier (2006), relies on cointegration assumptions and hence the use of structural vector error correction models. This approach has the advantage of taking full account of the non-stationary properties of the series and is able to disentangle the short-run from the long-run dynamics. To distinguish the surprise technology shock from the anticipated one they impose zero short-run and long-run restrictions. The main variables of their analysis are TFP and stock prices. The news shock is orthogonal to TFP but affects it in the long-run like a surprise technology shock.⁷ Beaudry and Portier (2014) argue that the news shock is the main driver of the business cycle, generating comovement among

⁶However, in her overview of the literature, Ramey (2016) shows that often the news shocks generated by estimated DSGE models are not unanticipated.

⁷For the systems that contain three or more variables the results from the original work of Beaudry and Portier (2006) cannot be replicated as pointed out by Kurmann and Mertens (2014). Following this critique, Beaudry and Portier (2014) propose a modification of their original identification strategy to extend it to a larger system, but do not provide a full identification for more than 3 variables.

output, consumption, investment and hours worked. The news shock accounts for more than half of the forecast error variance decomposition of these variables. However, the response of TFP to the news shock is not only delayed, but actually turns negative for the first two years, making the news shock rather unimportant for TFP at business cycle frequencies. Subsequent work from Beaudry and Lucke (2010) finds that the technology news shock is the dominant force, even when it competes with other technology or non-technology shocks.⁸ Fève and Guay (2018) obtain similar conclusions when they introduce sentiment shocks in a VEC model. Our paper does not find a major role for the technology news shock before the 80's. However, its relevance for the macroeconomic quantities from the 80's is accompanied by a high relevance for TFP as well.

The second approach has its origins in the work of Barsky and Sims (2011) and is based on the maximization of the forecast error variance decomposition of TFP at some horizon, together with the assumption of orthoganality to TFP innovations. The authors argue that the news shock is not the main driver of the business cycle, because it does not generate comovement among the main macroeconomic variables. Their methodology has been modified in various ways by the following literature, inspired also by Francis et al. (2014). Forni et al. (2014) and Kurmann and Sims (2017) partially confirm the original findings. Kurmann and Sims (2017) find that the technology news shock cannot be the main source of business cycle variations, as it does not generate positive comovement on impact between output and hours worked. However, it seems to account well for the forecast error variance decomposition at medium to longer frequencies.

Kurmann and Otrok (2013) and Görtz et al. (2016) find a large role for the technology news shock and show that credit spread indicators fall and the slope of the yield curve rises after a technology news shock, respectively. Gambetti et al. (2018) show that news shocks have a different effect on yields and spreads before and after 1980 and relate it to the conduct of monetary policy. They also argue that the news shock identified after the 1980 looks more similar to the one identified in Beaudry and Portier (2006). However, unlike us, they find a negative response of inflation to the news shock. Bouakez and Kemoe (2017) show that the identification strategy in Barsky and Sims (2011) is sensitive to the presence of measurement errors in TFP. These errors may explain why the identified surprise technology shocks induce an increase in inflation. This is important, since news shocks are identified to be orthogonal to the surprise shocks. Using a modified approach, Bouakez and Kemoe (2017) find a prominent role for the news shock, while the disinflationary puzzle disappears. Inflation falls after a surprise technology shock, while it is almost

⁸A comment from Fisher (2010) shows that the results in Beaudry and Lucke (2010) are sensitive to the cointegration assumptions. That would not have been a problem, if the number of cointegrated relationships of the system was not largely debatable from a theoretical standpoint.

unchanged after the news technology shock.

A contemporaneous study on the main driver of the business cycle is Angeletos et al. (2018). The authors use a methodology for the identification of the shocks, based on the maximization of the volatility of a macroeconomic variable at different horizons. The results point towards a non-inflationary demand shock as the main driver of the business cycle volatility, despite not being the most important for consumption. This main business cycle shock is not the most important driver of the long-run behaviour. On the contrary, the technology news shock we identify looks like a demand shock and it is the source of fluctuations at business-cycle frequencies and in the long run.

The paper is structured in the following way. Section 2 describes the data and its properties. Section 3 describes the vector error correction model and the identifying assumptions for the structural shocks. Sections 4 describes the dynamics after the identified shocks and their contribution to the business cycle. Section 5 discusses the propagation of the technology news shocks across different sample periods. Section 6 provides additional results and section 7 concludes.

2 Data and its characteristics

We base our analysis on a core vector error correction model with four variables. Later, we extend the model to additional variables to better understand the transmission of the identified shocks.

2.1 Core variables

For our analysis we utilize quarterly U.S. data for the period 1948:01-2017:03. The main variables of interest consist of a measure of total factor productivity, output, consumption and investment. To be able to identify technology shocks (anticipated and unanticipated ones) we need a measure of total factor productivity (TFP). We consider the widely used real-time measure of TFP produced by Fernald (2014), that corrects for capital utilization based on the methodology proposed by Basu et al. (2013). Since the correction is based on annual data, Fernald cannot reproduce the exact same correction with quarterly data, but takes into account the most important elements and this is why the measure is commonly used in the literature on technology news shocks.⁹ For the macroeconomic variables, we

⁹The series is continuously updated and available for download at https://www.frbsf.org/economic-research/indicators-data/total-factor-productivity-tfp/. We use the December 2017 vintage. Kurmann and Sims (2017) discuss the changes in the series, due to the revision in the utilization measurement carried out in March 2014, when Fernald switched from following Basu et al. (2006) to Basu et al. (2013). This revision motivated the comment by Cascaldi-Garcia (2017b) to Kurmann and Otrok (2013).

consider the real chain-weighted measures of GDP (output), personal consumption expenditure (consumption) and gross private domestic investment (investment) provided by the Bureau of Economic Analysis¹⁰ (BEA). We transform the series in per-capita terms by dividing them with the series of civilian noninstitutional population of 16 years and above provided by the Bureau of Labor Statistics (BLS). Finally, we consider the logarithmic transformation of all these variables.

2.2 Additional variables

The variables mentioned below enrich the core model in order to highlight the transmission of the shocks. The corporate spread corresponds to the difference between Moody's Baa and Aaa corporate bond yield and is available in FRED database.¹¹ It is available at monthly frequency and we transformed it to quarterly series by taking the average value for each quarter. Inflation is measured as the annual growth rate of the consumer price index, provided by BLS. Results are available also for an alternative measure usually considered in the literature, that is the annual growth rate of the GDP deflator (BLS). Using the annual rate of inflation, we can rule out any problem related to seasonal adjustment. The interest rate used is the 3-month Treasury bill yield, deflated with actual CPI inflation. The Treasury yield is available at monthly frequency and we take averages of each quarter to make it quarterly.

Hours worked refers to the nonfarm business sector and the measure is also provided by BLS. We transform it in per-capita terms with the same population series used above. Confidence is measured as the index of consumer sentiment from the Michigan Survey of Consumers, available only from 1960.¹² Uncertainty is measured as the second factor representing financial uncertainty in Jurado et al. (2015) and it is available only from 1960q3.¹³

2.3 Unit roots testing

Unit root tests¹⁴ cannot reject the hypothesis that the core variables are non-stationary, in particular they are unit roots with a drift. Based on these tests, the corporate spread, the interest rate, CPI inflation, confidence and uncertainty are stationary, while results are mixed for the growth rate of the GDP deflator and hours worked. Indeed, there is an ample

¹⁰More precisely, they correspond to Table 116.

¹¹FRED data are available for download at https://fred.stlouisfed.org/. We have considered also the spread measured as difference between the Baa corporate bond yield and the 10-year Treasury bond yield for robustness.

¹²The data are available for download here: http://www.sca.isr.umich.edu/.

¹³The data are available for download from one of the authors' website: https://www.sydneyludvigson.com/data-and-appendixes/.

¹⁴Results from the tests are not reported here but they are available from the authors upon request.

debate on whether hours worked is stationary or not and we will discuss the issue when we introduce the variable in our model.

3 Model and Identification

In this section we analyze the presence of cointegration among our core variables. We describe the vector error correction model used in the analysis and the assumptions behind the zero short and long-run restrictions used to identify the shocks. The choice of having this system as a core rests on the need to incorporate as much information as possible to avoid non-fundamentalness issues, while being parsimonious.

3.1 Cointegration

We have mentioned above that our main series of interest, TFP, output, consumption and investment, are unit roots. Now we study the presence of cointegration among them, implying that linear combinations of these variables may be stationary. It is indeed very plausible that output, consumption and investment co-move in the long-run in a stationary way, given the national account identity. In fact, King et al. (1991) have already exploited the presence of cointegration among them in their analysis. Here, we also consider the relationship among these variables and TFP. Hence, we expect to find two cointegrating relationships for the system of four variables. We use the Johansen trace test of cointegration. The test for lag selection suggests 3 lags. Since we use quarterly data with seasonally adjusted series, it is useful to have at least 4 lags, hence one full year of past data, to account for possible seasonality left in the series. Finally, we use 6 lags in order to rule out the presence of autocorrelation in the residuals. The results for cointegration are similar with 4 lags. We assume the presence of a linear trend in the long-run relationship, because this is the most general assumption about the deterministic components.¹⁵ The test, reported in Table 1, suggests the presence of two cointegrating relationships. Therefore, we proceed by assuming two cointegrating relationships and estimate a vector error correction model. As we will see, the trend enters the cointegrating vector in a significant way, hence we keep it.¹⁶

¹⁵See Juselius (2006) for the importance of the deterministic components for cointegration tests. The cointegration test relies on the assumption on the deterministic components and the inference about the deterministic components depends on the assumed cointegration. Hence, Juselius (2006) suggests to start from the most general case, i.e. the trend in the cointegrating vector, test for cointegration and then check the significance of the trend. If it is not significant, the second most-general assumption should be made about the determistic components and so on with the test.

¹⁶We have conducted a sensitivity analysis on the presence of the linear trend and found similar results for the identification of the news shock without it.

Given that our model contains the main macroeconomic variables and TFP, we find the results of the Johansen test in line with theory and proceed with two cointegrating relationships.

Cointegration test							
p-r	r	Eig.Value	Trace	Frac95	P-Value		
4	0	0.118	80.349	63.659	0.001		
3	1	0.102	46.063	42.770	0.022		
2	2	0.033	16.709	25.731	0.445		
1	3	0.027	7.552	12.448	0.299		

Table 1: Johansen trace test of cointegration for TFP, output, consumption and investment series with restricted trend and 6 lags. The table reports the eigenvalues, the trace statistic test, the 5% critical value and the p-value for the null hypothesis of r cointegrating relationships.

3.2 Vector Error Correction Model

Our four-variable vector error correction model (VECM) takes the following form.

$$\Delta X_t = \Pi X_{t-1}^* + \Gamma_1 \Delta X_{t-1} + \ldots + \Gamma_p \Delta X_{t-p} + \Theta D_t + \epsilon_t, \tag{1}$$

where X_t is the vector containing TFP, consumption, output and investment, X_{t-1}^* is the cointegrating vector containing the lagged variables and the deterministic terms, p is the lag chosen, D_t contains the deterministic terms outside the cointegrating vector. If it contains a constant term, it means that we have a linear trend in levels outside the cointegrating relationship. In case of cointegration, the matrix Π has reduced rank and can be decomposed as:

$$\Pi = \alpha \beta'. \tag{2}$$

As suggested by Johansen test above, it has rank 2, so α has size 4×2 , while β has size $(4+1) \times 2$, because we have included a trend in the cointegrating relationship. The vector β is the cointegrating vector and it defines the two stationary cointegrating relations, $\beta' X_{t-1}^*$. The vector α contains the loading coefficients, defining how the system adjusts to the long-term relationship among the variables. The other coefficient matrices of the VEC model define the short-term dynamics of the series.

Table 2 reports the loading coefficients for the two cointegrating relationships. We can see that TFP and investment are adjusting in a significant way to the long-run relationships, while consumption and output are weakly exogenous, since the coefficients in front of both

cointegrating vectors are not significant. Table 3 reports the estimates of the β vector, after normalization.¹⁷ All the variables enter significantly, including the linear trend, although with a small coefficient. A way to interpret the cointegrating vectors is to consider them in first difference. The first vector implies that the growth rate of output is proportional to the growth rate of consumption and investment plus a constant. The second vector implies that the growth rate of TFP and consumption plus a constant.¹⁸ We can interpret the linear trend as representing an additional variable missing from the system.

	$\alpha(1)$	$\alpha(2)$
DLTFP	-0.243	0.250
	(-3.773)	(4.085)
DLC	-0.045	0.010
	(-0.741)	(0.173)
DLY	0.063	0.051
	(0.938)	(0.793)
DLI	1.075	-0.514
	(3.828)	(-1.927)

Table 2: Loading coefficients for the model with TFP, output, consumption and investment series, with linear trend inside the two cointegrating vectors and 6 lags. Sample period is 1948-2017.

	LTFP	LC	LY	LI	Trend
$\beta'(1)$	0.000	1.000	-0.806	-0.079	-0.001
	(.NA)	(.NA)	(-13.201)	(-4.081)	(-3.970)
$\beta'(2)$	-0.126	1.000	-0.833	0.000	-0.001
	(-3.309)	(.NA)	(-13.651)	(.NA)	(-3.986)

Table 3: Cointegrating vectors for the model with TFP, output, consumption and investment series, with linear trend inside the cointegrating vectors and 6 lags. Sample period is 1948-2017.

Figure A.1 and A.2 in Appendix A report the graphs of the two cointegrating vectors. We can notice that they look stationary.

3.3 Identifying Assumptions

In order to structurally identify the VEC model, we introduce the Beveridge-Nelson MA representation of a *K*-dimension VEC model with *p* lags, derived by Johansen (1995).

$$X_{t} = \Xi\left(\sum_{i=1}^{t} u_{t} + \Theta t\right) + \sum_{j=0}^{\infty} \Xi_{j}^{*} L^{j} u_{t} + X_{0}^{*},$$
(3)

¹⁷Note that the normalization does not affect the results.

¹⁸But the growth rate of TFP is also related to the growth rate of investment by transitivity.

where $\Xi = \beta_{\perp} \left(\alpha'_{\perp} (I_K - \sum_{i=1}^{p-1} \Gamma_i) \beta_{\perp} \right)^{-1} \alpha'_{\perp}$ and has rank K - r. The element $\Xi \left(\sum_{i=1}^{t} u_t + \Theta t \right)$ is non-stationary and contains the long-run effects of the shocks and the deterministic components. The matrices $\Xi_j^{*'}$ s contain transitory effects and constitute the stationary component. The relationship between the original structural shocks ϵ_t and the reduced-form shocks u_t is given by:

$$u_t = B\varepsilon_t. \tag{4}$$

The matrix *B* is called the "impact matrix" and defines the short-run relationship between the structural shocks and the variables. The long-run effects of the shocks on the variables are defined by the long-run impact matrix, ΞB . In order to identify the responses of the variables to the structural shocks, we need to place some restrictions on one or both matrices.

Notice that for the full identification of a *K*-variables system, we need K(K-1)/2 restrictions. However, the assumption of cointegration implies that there are at least two permanent innovations in the system. The matrix of long-run effects contains only two linearly independent rows. Hence, each column of zeros in the long-run matrix ΞB represents only two (K - r) independent restrictions, requiring us to make additional assumptions in the impact matrix *B*.We consider the variables ordered in the following way:

$$X_t = \begin{bmatrix} TFP_t \\ Consumption_t \\ Output_t \\ Investment_t \end{bmatrix}$$

For our identification we use zero short-run and long-run restrictions. First, we distinguish between technology and non-technology shocks, by assuming that non-technology shocks cannot affect TFP in the long run. This is equivalent to imposing zero restrictions on the first element of the third and fourth column of matrix ΞB , so that the first two shocks are technology shocks. Zero long-run restrictions to identify shocks in VAR models were first proposed by Blanchard and Quah (1989) and applied to the context of technology shocks by Gali (1999). We further assume that only technology shocks can affect TFP in the short-run, imposing zero in the first elements of the third and fourth column of matrix *B*. To distinguish between surprise (first) and news technology shocks (second), we follow Beaudry and Portier (2006) and assume that only the surprise shock can affect TFP in the short run, imposing a zero in the first element of the second column of matrix *B*. This is in line with the idea that the technology news shock hits the economy before TFP changes

actually take place.

We are left with two non-technology shocks. Using Blanchard and Quah (1989)'s type of restrictions, we assume that the fourth shock is a temporary one and cannot affect investment in the long run. This implies a zero in the last element of the fourth column of matrix ΞB . As explained above, due to the assumption of two cointegrating relationships, two zeros in a column of matrix ΞB imply a column of zeros. Hence, our assumption is equivalent to assuming that the fourth shock has no long-run effect also on consumption and output. These assumptions imply the zero short and long-run restrictions reported below.

Therefore, ϵ_1 is the surprise technology shock, ϵ_2 is the news technology shock, ϵ_3 is a permanent non-technology shock, ϵ_4 is a temporary non-technology shock. We can summarize our assumptions in the following way.

Assumption 1. Only the surprise technology shock can affect TFP on impact and in the long run.

Assumption 2. The news technology shock is allowed to affect TFP only in the long run.

Assumption 3. *The non-technology shocks cannot affect TFP on impact or in the long run. Moreover, the fourth shock is temporary.*

When discussing the impulse response functions in the next section, we will show that the first two shocks indeed look like a technology surprise and news shock. Notice that the third shock is allowed to affect the macroeconomic quantities in the long run, while not being a technology shock. Later we will show that it looks like a labor supply shock, in the spirit of Shapiro and Watson (1988).

With respect to Beaudry and Portier (2006), we have replaced the variable of stock prices with the variable of consumption to identify news shocks. There are two reasons for this choice. First, the statistical properties of the series raise doubts about its inclusion in a VEC model. As discussed in Di Casola and Sichlimiris (2018), the series behaves like unit

root without drift up to the 80's and with drift from the 80's. Therefore, systems that contain stock prices need to account for a structural break to identify the cointegrating relationship in the data. The identification of news shocks in a VEC model with stock prices are sensitive to the data period used for the analysis. Such problems disappear when the consumption series is used instead. Second, the use of stock prices relies on the assumption that the stock market reacts positively on impact to the arrival of news on future technological improvements. As explained in Ramey (2016), this is not necessarily the case if the technological changes make current technology less valuable or happen in private companies not yet traded in the stock market. The idea of using consumption to study news shock was first proposed by Cochrane (1994), although he does not identify pritimive structural shocks. The measure of consumption aggregates the behaviour of all the consumers, who, individually, possess information about the future fundamentals that affect their life.

4 Dynamics and Contribution to the Business Cycle

In this section we document the dynamics induced by the identified two technology and two non-technology shocks on the main macroeconomic variables. After commenting on the dynamic behaviour of the shocks, we discuss their contribution for each macroeconomic variable based on the forecast error variance decomposition. The technology news shock is the dominant source of business cycle variation. We show that the news shock explains 60-70% of the unexplained variation in output, more than 70% of consumption and around 80% of investment. Importantly, it explains almost half of the unexplained variance of TFP within 40 quarters.

4.1 Impulse Responses from the core SVECM

Figure 1 displays the impulse responses from our core structural vector error correction model.

The (permanent) surprise technology shock

The first column of Figure 1 displays the responses of the surprise technology shock, that is the shock that can affect all the variables of the system both in the short and in the long run. The shock has a permanent effect on TFP and the macroeconomic variables. All the variables stabilize at a higher level after around two years. Even if this shock has a strong effect on TFP, it does not have a strong permanent effect on consumption, output and



Figure 1: Impulse response functions for the 4-variable VEC model with TFP (LTFP), consumption (LC), output (LY) and investment (LI), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the four shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

investment. In fact, we will see in section 5.3 that hours worked fall after this technology shock.

The (permanent) technology news shock

The second column of Figure 1 displays the impulse responses after the so-called news shock. As we have discussed previously, the technology news shock is identified following Beaudry and Portier (2006)'s identification paradigm. Even if this shock is not allowed to affect TFP on impact, it has a sizable effect on TFP starting from the eighth quarter. In the first eight quarters its effect on TFP is not significantly different from zero, in line with the idea of a delayed technological diffusion process, as theorized in Beaudry and Portier (2006).¹⁹ Due to the dynamics induced on TFP, we follow the naming convention

¹⁹Notice that the effect of the news shock on TFP in the first quarters is actually found to be negative and significant in Beaudry and Portier (2006), Beaudry and Portier (2014) and positive and significant in Barsky and Sims (2011). This is in contrast with the idea of a delayed response after the news shock.

of the literature on news shocks and we call it technology news shock. This shock has a permanent strong impact on all the macroeconomic variables included in our system. The response for all the variables is initially hump-shaped and then converges towards its long-run value. One may wonder why the effect of a surprise technology shock on the macroeconomic variables is not as strong as for the news shock, despite a strong effect on productivity. The answer lies on the propagation mechanism, as we show in section 5.

The permanent non-technology shock

The third shock in our system is a non-technology shock in the sense that it cannot affect TFP neither in the short run nor in the long run. On the other hand, it is allowed to have an effect on all other macroeconomic variables. The responses are displayed in the third column of Figure 1. This shock induces a permanent positive effect on all the macroeconomic quantities. However, it seems that the effect on output and consumption is stronger than the effect on investment. We have identified a non-technology shock that can induce positive correlation among the macroeconomic variables. In section 5.3 we discuss the propagation of this shock and argue that it can be interpreted as a labor supply shock in the spirit of Shapiro and Watson (1988).

The transitory non-technology shock

The last shock we identify from our core system is a shock that is not allowed to affect TFP neither in the short run nor in the long run. In addition, it can affect the macroeconomic quantities only in the short run. This shock has a temporary positive effect on output and investment, but reduces consumption mildly for the first quarters.

From the description of the impulse responses one can conclude that we have three shocks that can induce comovement in the macroeconomic quantities. Among these three shocks, the technology news shock stands out. In the next subsection, we document the forecast error variance decomposition after the four aforementioned shocks for TFP and the macroeconomic quantities.

4.2 Forecast Error Variance Decomposition from the core sVECM

The purpose of this section is to show that the news shock is the major source of variation of the core macroeconomic variables we have included in the system.

4.2.1 Technology News matter for TFP at Business Cycle Frequencies

The literature on technology news shocks that follows Beaudry and Portier (2006)'s paradigm has struggled to make technology news shocks important drivers of TFP at medium, business cycle frequencies. In fact, in their literature review on news shocks, Beaudry and Portier (2014) find that technology news shocks explain only 4% of the movements in TFP four years ahead. In Figure 2 we show that the technology news shock explains a significant part of the variation in TFP at the 5-10 years horizon (up to 45%). A similar result is obtained by Fève and Guay (2018). The rest of the movements in TFP are vastly explained by the surprise technology shock as one would expect, given that the other two shocks that we identify are non-TFP shocks.



Figure 2: Forecast error variance decomposition for the 4-variable VEC model with TFP, consumption, output and investment, including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the contribution of the surprise TFP shock (dotted orange), the TFP news shock (blue), the non-TFP permanent shock (striped grey) and the non-TFP transitory shock (yellow).

4.2.2 Technology News shocks: the main source of economic fluctuations in the short run and the long run

Having discussed the sources of variation in TFP, we show the sources of variation of the macroeconomic variables up to 40 quarters ahead. The technology news shock is responsi-

ble for most of the unexplained variation of the macroeconomic quantities at any frequency (Figure 2). At the 5-10 year horizon, the share due to the news shock ranges between 65% and 84% for output, consumption and investment. A smaller portion is explained by the permanent non-technology shock at any business cycle frequency. The surprise technology explains around 3% at any frequency. Finally, the transitory non-technology shock accounts for some movements in the very short run. A similar leading role for the technology news shock is found also in Beaudry and Lucke (2010), Beaudry and Portier (2014) and Fève and Guay (2018). This performance of our identified technology news shock is remarkable, given that it competes with two additional shocks that are allowed to have permanent effects on the real quantities.

5 On the propagation of the technology news shock

Up to now, we have shown that there are three shocks that can induce positive comovement in real economic activity: the two technology shocks and the permanent non-technology shock. The technology news shock explains the bulk of the variation in the macroeconomic quantities at any frequency. The permanent non-technology shock is the second most important one. In this section we extend our core system with a fifth variable in order to understand why the news shock is the leading source of business cycle fluctuations among these three candidates. Therefore, we place at the center of our analysis the propagation mechanism of the news shock and how that distinguishes it from the other shocks. The fifth variable will be related to CPI inflation, the financial conditions and the monetary policy response, one at a time.

From the estimated VEC model that contains inflation we show that the technology news shock behaves like a demand shock. Then, we show that the propagation of the technology news shocks was relatively weaker before the 80's due to the weaker financial transmission accompanied by relatively more persistent monetary policy response. To illustrate this argument, we estimate a VEC model that contains the corporate Baa spread and the real rate as a fifth variable respectively.

5.1 The inflationary technology news shock and the non-inflationary non-technology shock

The first variable we consider important to include in our system is inflation. The reason is that, inflation can inform us about the potential nature of the shock, whether it is a supply

shock or a demand shock.²⁰ The inflation (annual) rate is included in the vector error correction model like all other variables, but we restrict its coefficient in the cointegrated vector to be zero. Hence, we maintain the same cointegrating vectors as in the 4-variable system. Inflation is allowed to adjust to the long-run relationship among TFP, output, consumption and investment, but it is not allowed to affect it. Fève and Guay (2018) have followed a similar approach in their VEC model. Adding CPI inflation to the core system allows us to identify one more shock, which can serve as a robustness check for the results that we have already presented. The fifth shock is not allowed to affect any variable of the core system on impact, but is allowed to affect them in the long run.

$$B = \begin{bmatrix} * & 0 & 0 & 0 & 0 \\ * & * & * & * & 0 \\ * & * & * & * & 0 \\ * & * & * & * & * \end{bmatrix}$$
(7)

	*	*	0	0	*
	*	*	*	*	*
$\Xi B =$	*	*	*	*	*
	*	*	*	0	*
	*	*	*	*	*

We consider these identification assumptions conservative, because we allow for one more shock to impact all the variables permanently. In Figure 3 we show the responses of the variables after the five identified shocks.

After a surprise technology shock the response of inflation is negative for the first two quarters and then it becomes not significantly different from zero. Hence, the surprise technology shock looks indeed like a supply shock. The significance of our structural VEC model to identify well the surprise technology shock together with the technology news shock cannot be overlooked. In fact, Bouakez and Kemoe (2017) show that models based on Barsky and Sims (2011)'s identification of news shocks usually obtain a positive response of inflation after a surprise technology shock. The same is obtained in Gambetti et al. (2018) and Fève and Guay (2018). Given that the technology news shock is constructed to be orthogonal to the surprise technology shock, problems with the identification of the latter raise doubts about the identification of the former.

²⁰We call supply shock the type of shock that causes negative comovement in quantities and prices and demand shock the one that causes positive comovement in quantities and prices.



Impulse responses - extended sVECM with inflation

Figure 3: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and annualized inflation(INFL_ AN), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Inflation. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

After a news shock the response of inflation is weakly positive on impact, then it is hump-shaped and remains persistently positive. The literature on news shocks usually finds a negative response on impact and in the medium run (Barsky and Sims (2011), Görtz et al. (2016), Kurmann and Sims (2017), Gambetti et al. (2018)). In Bouakez and Kemoe (2017) this "disinflation puzzle" disappears and the response of inflation is not significant. The positive response we obtain indicates that our identified news shock behaves more like a demand shock rather than a supply shock. Finally, the second most important shock in the system is the permanent non-technology shock. The response of inflation after this shock is not significantly different from zero.

Due to the behavior of CPI inflation after a technology news shock and the permanent non-technology shock, we investigate whether these results are dependent on the measure of inflation that we use. Therefore, we replace inflation in the five variable VECM with the GDP deflator growth. We report these results in Figure B.1 in Appendix B. We confirm that the response of the GDP deflator growth is the same as for the CPI inflation after a technology news shock. However, the response is different for the non-technology permanent shock, becoming negative. This result implies that the third shock may be a supply shock that does not affect TFP in the long run, but affects output in the long run. At this point we conjecture that this shock may be the labor supply shock described in Shapiro and Watson (1988). In section 5.3 we report the response of hours worked and notice that the permanent non-technology shock increases hours worked and is its main driver in the long run.

We can conclude that the most important source of the business fluctuations is an inflationary technology news shocks and the second most important source is a non-technology non-inflationary shock.

5.2 When do technology news shocks matter?

In this section, we show that the financial transmission of the technology news shock is weak before the 80's. In addition, monetary policy does not restrain its propagation as much as before the 80's and compared to non-TFP shocks. To illustrate our argument, we consider a shorter sample period that spans from 1948:01 to 1979:01 in order to investigate different periods of the recent U.S. economic history. The choice of the end date for the short sample is not random. In July 1979 Volcker's era began as Volcker was appointed chairman of the Federal Reserve Board. The same month the U.S. Congress discussed the bill "Depository Institutions Deregulation and Monetary Control Act of 1980". This act gradually phased out restrictions for banks while also improving the Federal Reserve's control of monetary policy, by extending reserve requirements to depository institutions.²¹ These two events are considered pivotal for the U.S. economic history after World War II. Finally, we estimate our model up to 2008:03 and find no significant impact of the Great Recession on the dynamic response of the identified shocks.

The response of technology news shocks across the three subsamples.

We revisit the core system and we estimate it across the three different sample periods. We keep the number of cointegration relationships and the number of lags constant across different sample periods. In Figure 4 we plot the responses of the four macroeconomic variables after a technology news shock. The response of TFP is different for the short sample compared to the extended samples. For the short sample TFP increases and stabilizes at a higher level after 10 quarters. In the extended samples TFP keeps increasing,

²¹For more information on the act, see McNeill and Rechter (1980). Strahan (2003) discusses the impact of the act on the real economy, due to the improved functioning of the banking system.



Figure 4: Impulse response functions for the 4-variable VEC model with TFP (LTFP), consumption (LC), output (LY) and investment (LI), including 2 cointegrating relationships and 6 lags. In the rows we represent the variables and in the columns are the three different sample periods: 1948-1979q1, 1948-2008q3, 1948-2017q3. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

quarter by quarter. The response of the macroeconomic variables - consumption, output and investment - is similar across sample periods. To understand what drives the differences in the news shock, we look at the surprise technology shock and the permanent non-technology shock across different subsamples (see Figure B.2 in Appendix B). The response of the macroeconomic variables after the permanent non-technology shock does not change significantly. However, the strong response of the macroeconomic variables after a surprise technology shock that characterizes the short sample weakens substantially in the full sample. In Table 4 we display the forecast error variance decomposition of output for the full sample and the short sample. The bulk of the gains in forecast error variance decomposition for the technology news shock in the full sample comes from the weaker role of the surprise technology shock going from the short sample to the full sample.

Horizon	Surp	rise TFP	TFP	News	Non-TFP Perm		Non-TFP Trans	
	full	short	full	short	full	short	full	short
4	1,0	2,2	57,6	63,6	25,6	33,9	15,8	0,3
8	2,9	14,0	60,7	46,2	29,0	39,6	7,4	0,2
24	3,2	32,4	66,6	25,0	27,7	42,4	2,4	0,1
40	2,9	36,0	71,1	20,2	24,8	43,8	1,3	0,1

Table 4: Forecast error variance decomposition of output in 4-variable model with TFP, consumption, output and investment, including 2 cointegrating relationship and 6 lags. "Full" refers to the full sample period, 1948:01-2017:03. "Short" refers to the period 1948:01-1979:01.

We still have to answer the question on why anticipated technological improvements became the dominant force of business cycle. For this purpose, we investigate the transmission of the shocks across the different sub-samples. Two natural candidates are available: the response of monetary policy and the financial transmission. We argue that the financing conditions allowed a stronger transmission of anticipations about future technological changes to the real economy.

The role of monetary policy in the propagation of technology news shocks

We estimate the five-variable model with the measure of inflation that we have presented in the previous section for the different sub-samples. For the sample that ends before the Great Recession, we do not observe substantial differences in the dynamic response of the variables across different shocks compared to the full sample. However, for the short sample the response of inflation is significantly positive for the first 10 quarters after a permanent non-technology shock (Figure B.3 in Appendix B). In the full sample this was not the case. At the same time, the response of inflation after the technology news shock differs across subsamples (Figure 5). Inflation falls not only on impact, but also in the long-run in the short sample. Hence, the news shock up to 1979:01 behaves like a supply shock. Does this different response of inflation imply a substantially different response of monetary policy? To shed light on this question, we estimate a five-variable system that includes the core variables and a measure of real rate. The real rate is defined as the difference between the Treasury 3-month bill rate and current inflation.²² Using current inflation to deflate the nominal rate means that we are assuming that current inflation is a good predictor of next period's inflation. Since the next period for us is one quarter ahead, it is a reasonable assumption. The identifying assumptions for the fifth shock, the shock to the real rate, are the same as the ones used for inflation.

²²Our choice of Treasury bill instead of the Federal funds rate is driven by the larger data series availability, since the series of Fed funds rate starts in 1954. In our case losing 7 years of observations is really important, because we estimate a long-run relationship and need as long time series as possible.



Impulse responses after a TFP news shock - extended sVECM with inflation

Figure 5: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and annualized inflation(INFL_AN), including 2 cointegrating relationships. In the rows we represent the variables and in the columns are the three different sample periods: 1948-1979q1, 1948-2008q3, 1948-2017q3. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

From Figure 6 we see that the response of the real rate after a positive technology news shock is always positive on impact, but it is more persistent in the short sample, while it's short-lived in the second sample. Hence, in the short run the response of monetary policy to the news shock is equally contractionary, incentivizing a saving behaviour. Savings can become productive investment with a well-functioning financial intermediation. However, monetary policy has a different relation with inflation across the sample periods. In the short sample the increase in real rate coincides with a decrease in inflation, while in the full sample inflation increases. These different responses indicate that the monetary policy was more concerned with fighting inflationary pressures from the 80's rather than before, as argued in Clarida et al. (2000), Orphanides (2003). This result stands in contrast to the findings in Gambetti et al. (2018), where the Federal Reserve is found to be more restrictive towards inflation before the 80's rather than afterwards.

Figure B.4 and B.5 in Appendix B report the response of the real rate to all the shocks for the full and the short sample, respectively. The response to the surprise technology



Impulse responses after a TFP news shock - extended sVECM with real tbill

Figure 6: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and real treasury bill (RTBILL), including 2 cointegrating relationships. In the rows we represent the variables and in the columns are the three different sample periods: 1948-1979q1, 1948-2008q3, 1948-2017q3. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

shock is not significant in both cases. Instead, the response to a positive non-technology permanent shock is contractionary and looks more persistent from the 80's.

The role of financial deregulation for the propagation of technology news shocks.

Based on the response of the real rate in the short run after a technology news shock, we cannot find differences across different sample periods. However, we find that the monetary policy response is more persistent before the 80's after a technology news shock. Thus, it is difficult to argue that the monetary policy response can explain the weaker transmission of the news shock before the 80's. In the next paragraphs we show that the financial transmission channel was substantially weaker before the 80's and that might have contributed, in connection with the monetary policy response, to the relative unimportance of these shocks. To demonstrate our argument we extend the core system by including the corporate Baa spread variable. For the identification of the fifth shock we use the identi-fying assumptions used for inflation. These assumptions allow us to identify a financial shock that can potentially affect permanently all the variables in the system. The assumptions that a financial shock affects the macroeconomic variables with a delay is used also in Gilchrist and Zakrajšek (2012) and Walentin (2014).



Figure 7: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and Baa corporate spread (BAA), including 2 cointegrating relationships and 6 lags. In the rows we represent the variables and in the columns are the three different sample periods: 1948-1979q1, 1948-2008q3, 1948-2017q3. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

In Figure 7 we show that after a positive news shock the Baa corporate spread does not respond in a significant way in the short sample. Instead, for the extended sample periods the response of the corporate Baa spread on impact is negative and stays significantly negative for around 8 quarters. The negative response of the corporate spread indicates an improvement of the financing conditions. That would have allowed the technology news shocks to propagate better so as to have a larger effect on the real variables. It is worth pointing out that the response of the Baa spread turns to be insignificant at the point of the realization of technology news. Indeed, this is also the quarter when the real rate turns to be insignificant. All the important actions needed for the propagation of news shock happen before its realization. When we look at the dynamic response of the Baa spread after the permanent non-technology shock, we do not see differences across the two

samples (Figure B.6 and B.7 in Appendix B). That can explain why we have not found any differences in the responses to this shock in the core four variable system across different sub samples. Summing up, we argue that the technology news shock is the driving force of the business cycle mostly from the 80's, due to its stronger propagation through the financial markets.

Gambetti et al. (2018) studies the transmission of the news shock in two different subsamples. They find that the news shock causes an increase in interest rates and corporate spread before the 80's and a fall afterwards. Hence, a common finding between their paper and ours is that after a positive news shock financial conditions were more favorable from the 80's. However, we find a similar short-run increase in the real rate, independently of the sample period, while inflation decreases before and increases afterwards. Our proposed explanation for the different role of the news shock is the improved financial intermediation after the regulatory reforms of the 80's, instead of the different response of monetary policy that Gambetti et al. (2018) find.

5.3 The response of hours worked after technology shocks

The literature on news shock is divided regarding the response of hours worked after a news shock. Based on the original work by Beaudry and Portier (2006), hours worked should increase after a news shock, as found also in Beaudry and Lucke (2010). Another body of the literature on news shocks (Barsky and Sims (2011), Forni et al. (2014), Kurmann and Sims (2017)) finds that hours fall significantly on impact, thus pointing to the conclusion that the news shock can not be the main driver of the business cycle. We include hours worked as a fifth variable in the core system as done before. The fifth shock is not allowed to affect the TFP and real macroeconomic quantities in the short run, but can do it in the long run. Figure 8 shows that hours worked increase significantly on impact and remain positive after a positive news shock, independent of the the sample period. This positive response is in accordance with our message that the major business cycle shock is a shock to expectations about future technology and it behaves like a demand shock.

In the first column of Figure B.8 we see that the response of hours worked after a surprise technology shock is negative on impact and turns positive after few quarters, although insignificantly. There is a large literature (see Ramey (2016)) that discusses what is actually the "right" response of hours worked after a technology shock. In fact, the response depends often on whether hours worked is assumed to be stationary or not, which is also the topic of the debate. Lindé (2009) concludes that hours worked fall after a technology shock if the series enters the VAR system in first difference and increase if the series enters in levels. Canova et al. (2010) show that hours worked fall after a technology shock, when



Impulse responses after a TFP news shock - extended sVECM with Hours worked

Figure 8: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and hours worked (LH), including 2 cointegrating relationships and 6 lags. In the rows we represent the variables and in the columns are the three different sample periods: 1948-1979q1, 1948-2008q3, 1948-2017q3. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

the long cycles in the series are well accounted for. We confirm Francis et al. (2014)'s result with a different identification strategy, based on the maximization of the FEVD at a long but finite horizon. Hence, our result is in line with the recent literature on technology shocks and shows that we have correctly identified this shock. Finally, the response of hours worked after the second most important source of business cycle variations, the nontechnology non-inflationary shock, is positive on impact and has a persistent positive effect, similarly to the response after a news shock. Combining this results with the inflation response, the technology news shock behaves like a demand shock.

6 Additional results

In this section we report the results of additional analysis that gives more insight and serves as robustness of our findings. First, we introduce the so-called sentiment and uncertainty shocks. Second, we allow for the identification of a TFP measurement error shock, to capture possible measurement problems related to the adjusted TFP series. Third, we test for non-fundamentalness, that means testing for the likelihood that our core model contains enough information to let us recover the structural shocks. Overall, our results are robust to the inclusion of the additional shocks and do not present serious non-fundamentalness issues. Fourth, we show the choice of the sample period for the analysis may explain the lack of consensus on the role of the news shock in the literature. Finally, we discuss more in detail the possible nature of the permanent non-technology shock we have identified.

6.1 News, confidence and financial uncertainty

In the last years the research has focused on different types of shocks as drivers of the business cycle, that are uncorrelated to current fundamentals. One such case is the sentiment shock. Sentiment shocks represent changes in expectations about future developments that actually never materialize. Angeletos and La'O (2013) and Angeletos et al. (2018) are two prominent examples of models where sentiment shocks have a leading role of the business cycle, generating waves of optimism and pessimism unrelated to fundamentals. Barsky and Sims (2012) argue that news shocks are more important than noise or sunspot shocks. Fève and Guay (2018) have identified news and sentiment shocks in a VAR model, where the news shocks turn out to be the most important shocks.

A second type of shock unrelated to fundamentals is the uncertainty shock. This shock captures the exogenous change in the volatility of forecast errors of macroeconomic or financial aggregates.²³ Increases in uncertainty shocks are found to be contractionary for the economy in Jurado et al. (2015).

In order to assess the robustness of our results, we allow for the introduction of sentiment and uncertainty shocks, one at a time. We extend our core model to include a measure of confidence, like the one used in Fève and Guay (2018), or a measure of financial uncertainty to recover sentiment and uncertainty shocks. Notice that the series of confidence and uncertainty are only available from 1960, hence we compare the results with our core model estimated from 1960.

We introduce the aforementioned variables as done before with inflation, spread and the real rate. The variables enter the VEC system, but not the cointegrating vector. To identify the sentiment shock and the uncertainty shock we use the same restrictions used in the previous 5-variable systems. Both shocks are assumed to be orthogonal to TFP and the macroeconomic variables on impact. The restrictions are in line with Fève and Guay (2018) and Jurado et al. (2015). In fact, we find that the sentiment shock is the main driver

²³We consider this type of uncertainty shocks because we use the data by Jurado et al. (2015), but other interpretations of uncertainty exist.

of confidence (80% of FEVD after 40 quarters), while the uncertainty shock is the main driver of financial uncertainty (45% of FEVD after 40 quarters).



Impulse responses after a TFP news shock -Comparison between core sVECM and extend sVECM's with confidence and uncertainty.

Figure 9: Impulse response functions after a technology news shock for three different structural VEC models for the period 1960q3-2017q3. Core (column 1): 4-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and hours worked (LH), including 2 cointegrating relationships and 6 lags. Core sVECM extended with confidence (column 2). Core sVECM extended with uncertainty (column 3). Rows represent the variables. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Figure 9 reports the comparison of the response to the news shock for the core system, the model with confidence and the model with uncertainty.²⁴ We notice that confidence rises in the short run after a news shock, as found in Barsky and Sims (2011), Beaudry and Portier (2014) and Fève and Guay (2018). A positive shock to sentiment brings about a boom in the economy. On the other hand, uncertainty is not affected by the news shock in the short run, but increases in the long run, although not significantly. The effect of the news shock on the macroeconomic variables does not seem to change in the extended model. At the same time, the share of forecast error variance decomposition explained

²⁴Figure B.10 and B.11 in Appendix B show the responses of all variables to all shocks identified in the systems with confidence and uncertainty.

by the news shock is marginally affected. While the news shock explains around 40% of FEVD of output at the 10-year horizon in the core model for the period 1960-2017, the share decreases to 30% in the model with confidence and 33% in the model with uncertainty. The share of non-technology permanent shocks goes from around 50% to 42% in the extended models. The sentiment shocks explains 22% of the FEVD of output at the 10-year horizon, while the uncertainty shock has a share of 16%. Overall, the introduction of the sentiment shocks and uncertainty shocks has a moderate impact on the behaviour and the importance of the news shocks. Hence, our results are robust to the inclusion of those shocks.

6.2 News and TFP measurement error shocks

The key variable for the identification of technology shocks is the measure of TFP. The most widely accepted measure of adjusted TFP is the one proposed by Fernald (2014), although problems with the adjustment of the series for capital utilization have emerged recently. In this section we analyse how sensitive are our results to the possibility that adjusted TFP is not well measured.

Sims (2016), Cascaldi-Garcia (2017b) and Kurmann and Otrok (2017) show how different vintages of the measure of TFP by Fernald (2014) affect the identification of technology news shocks. In particular, the important change seems to be the correction of the measurement of adjusted TFP undertaken in March 2014 by Fernald. Bouakez and Kemoe (2017) and Kurmann and Sims (2017) have proposed modifications to the identification strategy of technology news shocks by Barsky and Sims (2011), that makes the results robust to measurement errors. Bouakez and Kemoe (2017) introduce an additional technology shock, to account for TFP measurement errors, while Kurmann and Sims (2017) modify the assumption that TFP is orthogonal to technology news shocks on impact.

We consider the following identification for the four-variable VEC model. We allow the fourth shock to affect TFP on impact but not in the long run. Moreover, due to the two cointegrating relationships, this shock is implicitly assumed not to affect all the macroe-conomic variables in the long run.

	[*	0	0	*
в —	*	*	*	*
D =	*	*	*	*
	*	*	*	*

In this way we have three technology shocks: the first one can affect TFP in the short run and the long run; the second one can affect TFP only in the long run; the third (that is the fourth in the system) can affect TFP only in the short run. Hence, now the last two shocks are both temporary and we have ruled out the possibility of a permanent non-technology shock.

Figure 10 reports the impulse responses of the model. The news shock still affects TFP in the long run, as much as the surprise technology shock, although the response of TFP to the technology news shock now turns negative for the first quarters. The news shock is still the main driver of the business cycle, generating comovement among consumption, output and investment and accounting for most of their forecast error variance decomposition. Hence, our results are robust to the presence of TFP measurement errors in the series of adjusted TFP.

6.3 Testing for non fundamentalness in the core sVECM

The work by Beaudry and Portier (2006) is based on a small-scale VEC model, with 2 and up to 4 variables. Due to the small size of the model, Forni et al. (2014) have raised the issue of non-fundamentalness (see also Beaudry and Portier (2014) for a review on the topic), arguing that the news shock cannot be recovered in a small-scale system. The problem of non-fundamentalness is related to the case when the econometrician has less information than the agents in the economy and thus cannot recover the structural shocks. This is more likely when the econometrician tries to estimate technology news shocks, because the economic agents may have more information about the future fundamentals. To overcome this problem, the identification of the news shock needs to rely on a variable that contains expectations about the future. We have used consumption and we have shown that it is indeed useful to recover the news shock. Here we will show that our VEC model does not suffer from serious non-fundamentalness issues.

We conduct the test of non-fundamentalness suggested by Forni et al. (2014) and further enriched by Beaudry et al. (2016). The test consists on regressing the shock of interest on the lags of some factors constructed from a large set of macroeconomic variables. If the shocks are orthogonal to the factors, the shock identified is fundamental. Moreover, Beaudry et



sVECM with a TFP measurement error shock

Figure 10: Impulse response functions for the 4-variable VEC model with TFP (LTFP), consumption (LC), output (LY) and investment (LI), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the four shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP transitory, Column 4: TFP measurement error. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

al. (2016) show that the R-squared of the regression contains information about how serious is the problem of non-fundamentalness. Applying the test to the shocks identified by Beaudry and Portier (2006), Beaudry et al. (2016) find that the tests reject the hypothesis of orthogonality for 1 and 4 lags of the factors, but the values of R-suqared are low, ranging from 0.03 to 0.21. Following Fève and Guay (2018), we use the eight Ludvigson-Ng macroeconomic factors, based on Ludvigson and Ng (2009)²⁵. Since the factors are available only from 1960, we estimate our VEC model for the same period for consistency reasons, although, as discuss below, a longer sample period is important for the long-run relationship we estimate. We consider our core four-variable VEC model and the model augmented with inflation. Regressing the identified news shocks on the lags of the factors, we find that we reject the hypothesis of orthogonality (based on the F-test) at 5% level for the first lag of the factors. On the other hand, we cannot reject the hypothesis of orthogonality

²⁵The data are available for download at https://www.sydneyludvigson.com/data-and-appendixes/.

at 5% level for more lags. Moreover, the R-squared of the regression with the first lag is 0.095. Similar results, albeit with smaller R-squared (0.084) are obtained for the news shock identified with the 5-variable VEC model. Hence, the problem of nonfundamentalness is not a significant issue in our model.

6.4 Reconciling different views about the news shock: On the sample period choice

The literature on news shocks has not managed to agree on whether the news shock is the main driver of the business cycle (see Beaudry and Portier (2014) and Ramey (2016) for discussion). The problem can be summarized as follows. The part of the literature on news shocks that follows Beaudry and Portier (2006)'s paradigm argues that technology news shocks are the driving force of the business cycle, because they induce positive comovement among the real macroeconomic quantities and hours worked. Another strand of the literature, mostly based on Barsky and Sims (2011)'s approach, finds that news shocks cannot be the driver of the business cycle because they do not generate positive comovement in the main macroeconomic variables.²⁶ We show that the choice of the sample period might have played an important role for the differences between the two seemingly contradictory findings. Once we estimate our model for the sample period most used in the second strand of the literature, we obtain a different role for the news shock. Following Barsky and Sims (2011), most of the following literature start the analysis in the first quarter of 1960. We estimate the core model for the period 1960:01-2007:04.

In Figure 11 we notice that the response of the macroeconomic variables after a technology news shock is not as strong as before. In fact, even the output response is close to zero. Moreover, TFP responds positively to the technology news shock directly after the first quarter, as in Barsky and Sims (2011) and not in line with the theory of a delayed diffusion process. Looking at the forecast error variance decomposition, the technology news shocks is not the most important driver of the business cycle. However, to obtain such a result we have excluded the period before the 60's, that is a period of generally low growth, and the period during and after the Great Recession, that is again a period of low growth. Such sample selection implies that we are overestimating the potential growth rate of the U.S. economy with our VEC model. The implications from this can be seen in the historical decomposition of the level of output (Figure 12). If one believes that 1960-2007 is the "right" sample period, one has to accept that for almost the whole 80's the U.S. economy was operating far below its potential output. At the same time, the potential TFP

²⁶Additionally, Ramey (2016) shows that the news shocks identified with different approaches are basically uncorrelated with each other.



Figure 11: Impulse response functions for the 4-variable VEC model with TFP (LTFP), consumption (LC), output (LY) and investment (LI), including 2 cointegrating relationships and 6 lags. Sample period is 1960-2007. The graph displays the variables responses (rows) to the four shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

growth does not change so much going from the shorter to the full sample. This upward bias in GDP growth potential compared to TFP growth potential affects mostly the shock that links future technology changes with macroeconomic quantities. Hence, the shorter sample gives more importance to a non-technology force of the business cycle. That might explain why other studies find other shocks as driving forces. From our point of view, the longer sample period allows to better estimate the long-run relationship among the core variables.

Our result is not surprising and it actually complements the findings in Cascaldi-Garcia (2017a). Based on Barsky and Sims (2011)'s identification of news shocks, he finds that the response of the macroeconomic variables after a technology news shock is sensitive to the sample period chosen. He shows that the impact response of hours worked even switches sign. Based on our identification, the impact response of hours worked and other variables does not switch sign, but it is weaker than in the full sample. However, as we have argued,

Historical decomposition: Output - Barsky and Sims (2011) sample period



Figure 12: Historical decomposition of output from the 4-variable VEC model with TFP, consumption, output and investment, including 2 cointegrating relationships and 6 lags. Sample period is 1960-2007. The black line represents the variation from the baseline projection and the grey area is the contribution of the news shock.

one should trust more the results coming from the longer sample rather than the shorter one, especially for cointegrated systems.

6.5 Can the permanent non-technology shock be a labor supply shock?

In table 4 we have shown that the vast majority of business cycle variation for output for the post World War II period can be accounted for by the technology news shock. However, a small but non negligible share of the unexplained variation in output is accounted for by the non-technology shock that has a permanent effect on output. In this section we shed light on the nature of this shock.

The third shock of our core model is identified as a non-technology shock because it cannot affect technology neither in the short run, nor in the long run. However, it is distinguished from the fourth shock because it can affect output, consumption and investment at any horizon. Indeed, we have found that a positive non-technology permanent shock affects positively output in the long run (Figure 1), has a non-significant impact on inflation (Figure 3), but it decreases the GDP deflator growth (Figure B.1). Moreover, it increases the real Tbill rate (Figure B.4) and hours worked (Figure B.8).

The permanent non-technology shock is not the main driver of the business cycle in the full sample period, but it seems to have been as important as the surprise technology shock and the technology news shock before the 80's (Figure B.2). Up to the 80's, this shock explains almost all the variability of hours worked in the long run (Figure B.9). We conjecture that our identified non-technology permanent shock looks like the labor supply shock in Shapiro and Watson (1988). They identify the labor supply shock as the only shock

that can affect both output and hours in the long run, while the surprise technology shock can affect only output in the long run. They find that the labor supply shock is the main driver of the business cycle for the period 1951-1987. Unlike Shapiro and Watson (1988), we have identified the labor supply shock without imposing restrictions on hours worked, but instead imposing restrictions on technology (TFP). Hence, we have identified it indirectly and found similar results in a similar sample period. The results in the extended sample, though, show that the labor supply shock has lost importance for the business cycle.

7 Conclusion

This paper argues that the main source of business cycle fluctuations is a technology news shock that acts like a demand shock. Anticipated shifts in technology generate strong positive correlation between output, consumption and investment, and a weak positive correlation between these quantities and inflation. The technology news shock has become the most important driver of the business cycle from the 80's, coinciding with the era of financial deregulation. This shock was as important as the unanticipated technology shock and a non-technology permanent shock before the 80's. We attribute the relative increased importance of the technology news to the improved propagation through the financial sector. The response of corporate spread to the technology news shock is null before the 80's and negative afterwards, implying easing of financing conditions. Our preferred explanation is that the financial deregulation of the 80's improved the financial transmission of the technology news shock. At the same time, the monetary policy response was contractionary, no matter the sample period. The response of the real rate to the technology news shock from the 80's, though, is less persistent than before the 80's and with respect to the response to other non-technology shocks.

For our analysis we have utilized vector error correction models, identifying two technology and two non-technology shocks with zero short and long-run restrictions for our baseline model. The identifying assumptions for the technology news shock are based on the prominent work by Beaudry and Portier (2006), but we modify and enrich it in important dimensions. We use the measure of consumption instead of stock prices to identify news shocks. Moreover, we extend their model to have enough information as to overcome nonfundamentalness problems and identify other shocks that can compete with technology shocks.

Our identified technology news shock is in line with the benchmark New Keynesian model, because it induces an increase in economic activity and an increase, albeit weak, in inflation. However, to reproduce the rich dynamic responses of the variables, one may

need to incorporate additional features in the baseline model. Moreover, the identified role of the financial propagation of the shock provides support for DSGE models with financial amplification channels, like the one estimated in Görtz and Tsoukalas (2017).

References

- Angeletos, George-Marios and Jennifer La'O, "Sentiments," *Econometrica*, 2013, 81 (2), 739–779.
- _ , Fabrice Collard, and Harris Dellas, "Business Cycle Anatomy," NBER Working Papers 24875, National Bureau of Economic Research July 2018.
- Barsky, R. B. and E. R. Sims, "News shocks and business cycles," *Journal of Monetary Economics*, 2011, 58 (3), 273–289.
- Barsky, Robert B. and Eric R. Sims, "Information, Animal Spirits, and the Meaning of Innovations in Consumer Confidence," *American Economic Review*, June 2012, 102 (4), 1343–77.
- _, Susanto Basu, and Keyoung Lee, "Whither News Shocks?," NBER Macroeconomics Annual, 2015, 29 (1), 225–264.
- Basu, S., J. G. Fernald, and M. S. Kimball, "Are Technology Improvements Contractionary?," *American Economic Review*, December 2006, *96* (5), 1418–1448.
- **Basu, Susanto, John Fernald, Jonas Fisher, and Miles Kimball**, "Sector-Specific Technical Change," Mimeo, Federal Reserve Bank of San Francisco 2013.
- Beaudry, P. and F. Portier, "Stock Prices, News, and Economic Fluctuations," *American Economic Review*, September 2006, *96* (4), 1293–1307.
- **Beaudry, Paul and Bernd Lucke**, "Letting Different Views about Business Cycles Compete," in "NBER Macroeconomics Annual 2009, Volume 24" NBER Chapters, National Bureau of Economic Research, Inc, October 2010, pp. 413–455.
- and Franck Portier, "News-Driven Business Cycles: Insights and Challenges," *Journal of Economic Literature*, 2014, 52 (4), 993–1074.
- __, Patrick Fève, Alain Guay, and Franck Portier, "When is Nonfundamentalness in SVARs A Real Problem?," TSE Working Papers 16-738, Toulouse School of Economics (TSE) November 2016.
- Blanchard, O. J. and D. Quah, "The Dynamic Effects of Aggregate Demand and Supply Disturbances," *American Economic Review*, September 1989, 79 (4), 655–73.
- **Bouakez, Hafedh and Laurent Kemoe**, "News Shocks, Business Cycles, and the Disinflation Puzzle," Mimeo 2017.

- **Canova, F., D. Lopez-Salido, and C. Michelacci**, "The effects of technology shocks on hours and output: a robustness analysis," *Journal of Applied Econometrics*, 2010, 25 (5), 755–773.
- **Cascaldi-Garcia**, **Danilo**, "Amplification effects of news shocks through uncertainty," Technical Report, Mimeo 2017.
- ____, "News Shocks and the Slope of the Term Structure of Interest Rates: Comment," American Economic Review, October 2017, 107 (10), 3243–49.
- **Clarida, Richard, Jordi Galí, and Mark Gertler**, "Monetary policy rules and macroeconomic stability: evidence and some theory," *The Quarterly journal of economics*, 2000, 115 (1), 147–180.
- **Cochrane, John H.**, "Shocks," *Carnegie-Rochester Conference Series on Public Policy*, December 1994, 41 (1), 295–364.
- **Di Casola, Paola and Spyridon Sichlimiris**, "Stock prices, news and economic fluctuations: Comment," Mimeo 2018.
- **Fernald, J.**, "A quarterly, utilization-adjusted series on total factor productivity," Working Paper Series 2012-19, Federal Reserve Bank of San Francisco 2014.
- Fève, Patrick and Alain Guay, "Sentiments in SVARs," The Economic Journal, 2018.
- Fisher, Jonas D. M., "Comment on "Letting Different Views about Business Cycles Compete"," in "NBER Macroeconomics Annual 2009, Volume 24" NBER Chapters, National Bureau of Economic Research, Inc, October 2010, pp. 457–474.
- Forni, Mario, Luca Gambetti, and Luca Sala, "No News in Business Cycles," *Economic Journal*, December 2014, 124 (581), 1168–1191.
- **Francis, Neville, Michael T Owyang, Jennifer E Roush, and Riccardo DiCecio**, "A flexible finite-horizon alternative to long-run restrictions with an application to technology shocks," *Review of Economics and Statistics*, 2014, *96* (4), 638–647.
- Fujiwara, Ippei, Yasuo Hirose, and Mototsugu Shintani, "Can news be a major source of aggregate fluctuations? A Bayesian DSGE approach," *Journal of Money, Credit and Banking*, 2011, 43 (1), 1–29.
- Galì, Jordi, "Technology, Employment, and the Business Cycle: Do Technology Shocks Explain Aggregate Fluctuations?," *American Economic Review*, March 1999, *89* (1), 249– 271.

- Gambetti, Luca, Dimitris Korobilis, John D. Tsoukalas, and Francesco Zanetti, "The Effect of News Shocks and Monetary Policy," Working Paper series 18-19, Rimini Centre for Economic Analysis May 2018.
- Gilchrist, Simon and Egon Zakrajšek, "Credit Spreads and Business Cycle Fluctuations," *American Economic Review*, June 2012, *102* (4), 1692–1720.
- Görtz, Christoph and John D. Tsoukalas, "News and Financial Intermediation in Aggregate Fluctuations," *The Review of Economics and Statistics*, July 2017, 99 (3), 514–530.
- _ , _ , and Francesco Zanetti, "News Shocks under Financial Frictions," Working Papers 2016₁5, Business School Economics, University of Glasgow June 2016.
- Jinnai, Ryo, "News shocks and inflation," Economics Letters, 2013, 119 (2), 176 179.
- Johansen, S., "Identifying restrictions of linear equations with applications to simultaneous equations and cointegration," *Journal of Econometrics*, 1995, 69 (1), 111 132. Annals of Econometrics: Bayesian and Classical Econometric Modelling of Time Series.
- Jurado, Kyle, Sydney C Ludvigson, and Serena Ng, "Measuring uncertainty," American Economic Review, 2015, 105 (3), 1177–1216.
- **Juselius, K.**, *The Cointegrated VAR Model: Methodology and Applications* Advanced Texts in Econometrics, OUP Oxford, 2006.
- Khan, Hashmat and John Tsoukalas, "The quantitative importance of news shocks in estimated DSGE models," *Journal of Money, Credit and Banking*, 2012, 44 (8), 1535–1561.
- King, Robert G., Charles I. Plosser, James H. Stock, and Mark W. Watson, "Stochastic Trends and Economic Fluctuations," *American Economic Review*, September 1991, 81 (4), 819–40.
- Kurmann, A. and C. Otrok, "News Shocks and the Slope of the Term Structure of Interest Rates," *American Economic Review*, October 2013, *103* (6), 2612–32.
- _ and _ , "News Shocks and Inflation: Lessons for New Keynesians," Mimeo 2014.
- _ and _ , "News Shocks and the Slope of the Term Structure of Interest Rates: Reply," American Economic Review, October 2017, 107 (10), 3250–3256.
- _ and E. Mertens, "Stock Prices, News, and Economic Fluctuations: Comment," American Economic Review, April 2014, 104 (4), 1439–45.

- and E. Sims, "Revisions in Utilization-Adjusted TFP and Robust Identification of News Shocks," NBER Working Papers 23142, National Bureau of Economic Research, Inc February 2017.
- Lindé, Jesper, "The effects of permanent technology shocks on hours: Can the RBC-model fit the VAR evidence?," *Journal of Economic Dynamics and Control*, March 2009, 33 (3), 597–613.
- Lorenzoni, Guido, "News and Aggregate Demand Shocks," *Annual Review of Economics*, 09 2011, 3 (1), 537–557.
- Ludvigson, Sydney C and Serena Ng, "Macro factors in bond risk premia," *The Review of Financial Studies*, 2009, 22 (12), 5027–5067.
- McNeill, Charles R. and Denise M. Rechter, "The Depository Institutions Deregulation and Monetary Control Act of 1980," Federal Reserve Bulletin n.6 1980.
- **Orphanides, Athanasios**, "The quest for prosperity without inflation," *Journal of Monetary Economics*, April 2003, *50* (3), 633–663.
- Ramey, V. A., Macroeconomic Shocks and Their Propagation, Vol. 2 of Handbook of Macroeconomics, Elsevier,
- Schmitt-Grohé, Stephanie and Martin Uribe, "What's news in business cycles," *Econometrica*, 2012, *80* (6), 2733–2764.
- Shapiro, Matthew D. and Mark W. Watson, "Sources of business cycle fluctuations," NBER Macroeconomics annual, 1988, 3, 111–148.
- Sims, Eric R., "Differences in Quarterly Utilization-Adjusted TFP by Vintage, with an Application to News Shocks," NBER Working Papers, National Bureau of Economic Research, Inc April 2016.
- **Strahan, P.**, "The real effects of US banking deregulation," Federal Reserve Bank of St. Louis Review n.4 2003.
- Walentin, Karl, "Business cycle implications of mortgage spreads," *Journal of Monetary Economics*, 2014, 67 (C), 62–77.

Appendix A Cointegrating vectors



Figure A.1: Graphs of the first cointegrating vector for the model with TFP, output, consumption and investment series, with linear trend inside the cointegrating vector and 6 lags. The graph above reports the equilibrium error as a function of short-run dynamics and deterministic terms. The graph below reports the equilibrium error corrected for the short-run effects.



Figure A.2: Graphs of the second cointegrating vector for the model with TFP, output, consumption and investment series, with linear trend inside the cointegrating vector and 6 lags. The graph above reports the equilibrium error as a function of short-run dynamics and deterministic terms. The graph below reports the equilibrium error corrected for the short-run effects.

Appendix B Impulse Responses for the 5 variable VECM's

GDP deflator



Impulse responses - extended sVECM with GDP deflator growth

Figure B.1: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and GDP deflator growth(DGDPDEFL_ ANN), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: GDP defl Infl. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)



Figure B.2: Impulse response functions for the 4-variable VEC model with TFP (LTFP), consumption (LC), output (LY) and investment (LI), including 2 cointegrating relationships and 6 lags. Sample period is 1948:01-1979:01. The graph displays the variables responses (rows) to the four shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

CPI inflation



Figure B.3: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and annualized inflation(INFL_AN), including 2 cointegrating relationships and 6 lags. Sample period is 1948-1979:01. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Inflation. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Real Treasury bill yield



Impulse responses - extended sVECM with real t-bill

Figure B.4: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and real treasury bill yield (RTBILL), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Real t-bill. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Real Treasury bill yield



Impulse responses - extended sVECM with real t-bill - short sample

Figure B.5: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and real treasury bill yield (RTBILL), including 2 cointegrating relationships and 6 lags. Sample period is 1948-1979:1. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Real t-bill. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Baa corporate spread



Impulse responses - extended sVECM with corporate Baa spread

Figure B.6: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and corporate Baa spread (BAA), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Baa spread. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Baa corporate spread



Impulse responses - extended sVECM with corporate Baa spread - short sample

Figure B.7: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and corporate Baa spread (BAA), including 2 cointegrating relationships and 6 lags. Sample period is 1948-1979:01. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Baa spread. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Hours worked



Figure B.8: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and Hours worked(LH), including 2 cointegrating relationships and 6 lags. Sample period is 1948-2017. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Hours worked. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Hours worked



Impulse responses - extended sVECM with Hours worked

Figure B.9: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and Hours worked(LH), including 2 cointegrating relationships and 6 lags. Sample period is 1948:01-1979:01. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Hours worked. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Confidence



Figure B.10: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and Consumer confidence (CONF), including 2 cointegrating relationships and 6 lags. Sample period is 1960:3-2017:3. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: Confidence. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Uncertainty



Impulse responses - extended sVECM with Financial uncertainty

Figure B.11: Impulse response functions for the 5-variable VEC model with TFP (LTFP), consumption (LC), output (LY), investment (LI) and financial uncertainty (FIN_ UNC2), including 2 cointegrating relationships and 6 lags. Sample period is 1960:3-2017:3. The graph displays the variables responses (rows) to the five shocks: Column 1: surprise TFP, Column 2: TFP news, Column 3: non-TFP permanent, Column 4: non-TFP transitory, Column 5: financial uncertainty. We obtain the Bayesian simulated distribution of IRF by Monte Carlo integration with 5000 replications. The black solid line represent the median response. We plot the distribution of IRF at 16% to 84% interval (dark blue) and at 5% to 95% interval (light blue)

Earlier Working Papers:

For a complete list of Working Papers published by Sveriges Riksbank, see www.riksbank.se

Estimation of an Adaptive Stock Market Model with Heterogeneous Agents by Henrik Amilon	2005:177
Some Further Evidence on Interest-Rate Smoothing: The Role of Measurement Errors in the Output Gap by Mikael Apel and Per Jansson	2005:178
Bayesian Estimation of an Open Economy DSGE Model with Incomplete Pass-Through by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:179
Are Constant Interest Rate Forecasts Modest Interventions? Evidence from an Estimated Open Economy DSGE Model of the Euro Area by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:180
Inference in Vector Autoregressive Models with an Informative Prior on the Steady State by Mattias Villani	2005:181
Bank Mergers, Competition and Liquidity by Elena Carletti, Philipp Hartmann and Giancarlo Spagnolo	2005:182
Testing Near-Rationality using Detailed Survey Data by Michael F. Bryan and Stefan Palmqvist	2005:183
Exploring Interactions between Real Activity and the Financial Stance by Tor Jacobson, Jesper Lindé and Kasper Roszbach	2005:184
Two-Sided Network Effects, Bank Interchange Fees, and the Allocation of Fixed Costs by Mats A. Bergman	2005:185
Trade Deficits in the Baltic States: How Long Will the Party Last? by Rudolfs Bems and Kristian Jönsson	2005:186
Real Exchange Rate and Consumption Fluctuations follwing Trade Liberalization by Kristian Jönsson	2005:187
Modern Forecasting Models in Action: Improving Macroeconomic Analyses at Central Banks by Malin Adolfson, Michael K. Andersson, Jesper Lindé, Mattias Villani and Anders Vredin	2005:188
Bayesian Inference of General Linear Restrictions on the Cointegration Space by Mattias Villani	2005:189
Forecasting Performance of an Open Economy Dynamic Stochastic General Equilibrium Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2005:190
Forecast Combination and Model Averaging using Predictive Measures by Jana Eklund and Sune Karlsson	2005:191
Swedish Intervention and the Krona Float, 1993-2002 by Owen F. Humpage and Javiera Ragnartz	2006:192
A Simultaneous Model of the Swedish Krona, the US Dollar and the Euro by Hans Lindblad and Peter Sellin	2006:193
Testing Theories of Job Creation: Does Supply Create Its Own Demand? by Mikael Carlsson, Stefan Eriksson and Nils Gottfries	2006:194
Down or Out: Assessing The Welfare Costs of Household Investment Mistakes by Laurent E. Calvet, John Y. Campbell and Paolo Sodini	2006:195
Efficient Bayesian Inference for Multiple Change-Point and Mixture Innovation Models by Paolo Giordani and Robert Kohn	2006:196
Derivation and Estimation of a New Keynesian Phillips Curve in a Small Open Economy by Karolina Holmberg	2006:197
Technology Shocks and the Labour-Input Response: Evidence from Firm-Level Data by Mikael Carlsson and Jon Smedsaas	2006:198
Monetary Policy and Staggered Wage Bargaining when Prices are Sticky by Mikael Carlsson and Andreas Westermark	2006:199
The Swedish External Position and the Krona by Philip R. Lane	2006:200

Price Setting Transactions and the Role of Denominating Currency in FX Markets by Richard Friberg and Fredrik Wilander	2007:201
The geography of asset holdings: Evidence from Sweden by Nicolas Coeurdacier and Philippe Martin	2007:202
Evaluating An Estimated New Keynesian Small Open Economy Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Mattias Villani	2007:203
The Use of Cash and the Size of the Shadow Economy in Sweden by Gabriela Guibourg and Björn Segendorf	2007:204
Bank supervision Russian style: Evidence of conflicts between micro- and macro-prudential concerns by Sophie Claeys and Koen Schoors	2007:205
Optimal Monetary Policy under Downward Nominal Wage Rigidity by Mikael Carlsson and Andreas Westermark	2007:206
Financial Structure, Managerial Compensation and Monitoring by Vittoria Cerasi and Sonja Daltung	2007:207
Financial Frictions, Investment and Tobin's q by Guido Lorenzoni and Karl Walentin	2007:208
Sticky Information vs Sticky Prices: A Horse Race in a DSGE Framework by Mathias Trabandt	2007:209
Acquisition versus greenfield: The impact of the mode of foreign bank entry on information and bank lending rates by Sophie Claeys and Christa Hainz	2007:210
Nonparametric Regression Density Estimation Using Smoothly Varying Normal Mixtures by Mattias Villani, Robert Kohn and Paolo Giordani	2007:211
The Costs of Paying – Private and Social Costs of Cash and Card by Mats Bergman, Gabriella Guibourg and Björn Segendorf	2007:212
Using a New Open Economy Macroeconomics model to make real nominal exchange rate forecasts by Peter Sellin	2007:213
Introducing Financial Frictions and Unemployment into a Small Open Economy Model by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2007:214
Earnings Inequality and the Equity Premium by Karl Walentin	2007:215
Bayesian forecast combination for VAR models by Michael K. Andersson and Sune Karlsson	2007:216
Do Central Banks React to House Prices? by Daria Finocchiaro and Virginia Queijo von Heideken	2007:217
The Riksbank's Forecasting Performance by Michael K. Andersson, Gustav Karlsson and Josef Svensson	2007:218
Macroeconomic Impact on Expected Default Freqency by Per Åsberg and Hovick Shahnazarian	2008:219
Monetary Policy Regimes and the Volatility of Long-Term Interest Rates by Virginia Queijo von Heideken	2008:220
Governing the Governors: A Clinical Study of Central Banks by Lars Frisell, Kasper Roszbach and Giancarlo Spagnolo	2008:221
The Monetary Policy Decision-Making Process and the Term Structure of Interest Rates by Hans Dillén	2008:222
How Important are Financial Frictions in the US and the Euro Area by Virginia Queijo von Heideken	2008:223
Block Kalman filtering for large-scale DSGE models by Ingvar Strid and Karl Walentin	2008:224
Optimal Monetary Policy in an Operational Medium-Sized DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2008:225
Firm Default and Aggregate Fluctuations by Tor Jacobson, Rikard Kindell, Jesper Lindé and Kasper Roszbach	2008:226
Re-Evaluating Swedish Membership in EMU: Evidence from an Estimated Model by Ulf Söderström	2008:227

The Effect of Cash Flow on Investment: An Empirical Test of the Balance Sheet Channel by Ola Melander	2009:228
Expectation Driven Business Cycles with Limited Enforcement by Karl Walentin	2009:229
Effects of Organizational Change on Firm Productivity by Christina Håkanson	2009:230
Evaluating Microfoundations for Aggregate Price Rigidities: Evidence from Matched Firm-Level Data on Product Prices and Unit Labor Cost by Mikael Carlsson and Oskar Nordström Skans	2009:231
Monetary Policy Trade-Offs in an Estimated Open-Economy DSGE Model by Malin Adolfson, Stefan Laséen, Jesper Lindé and Lars E. O. Svensson	2009:232
Flexible Modeling of Conditional Distributions Using Smooth Mixtures of Asymmetric Student T Densities by Feng Li, Mattias Villani and Robert Kohn	2009:233
Forecasting Macroeconomic Time Series with Locally Adaptive Signal Extraction by Paolo Giordani and Mattias Villani	2009:234
Evaluating Monetary Policy by Lars E. O. Svensson	2009:235
Risk Premiums and Macroeconomic Dynamics in a Heterogeneous Agent Model by Ferre De Graeve, Maarten Dossche, Marina Emiris, Henri Sneessens and Raf Wouters	2010:236
Picking the Brains of MPC Members by Mikael Apel, Carl Andreas Claussen and Petra Lennartsdotter	2010:237
Involuntary Unemployment and the Business Cycle by Lawrence J. Christiano, Mathias Trabandt and Karl Walentin	2010:238
Housing collateral and the monetary transmission mechanism by Karl Walentin and Peter Sellin	2010:239
The Discursive Dilemma in Monetary Policy by Carl Andreas Claussen and Øistein Røisland	2010:240
Monetary Regime Change and Business Cycles by Vasco Cúrdia and Daria Finocchiaro	2010:241
Bayesian Inference in Structural Second-Price common Value Auctions by Bertil Wegmann and Mattias Villani	2010:242
Equilibrium asset prices and the wealth distribution with inattentive consumers by Daria Finocchiaro	2010:243
Identifying VARs through Heterogeneity: An Application to Bank Runs by Ferre De Graeve and Alexei Karas	2010:244
Modeling Conditional Densities Using Finite Smooth Mixtures by Feng Li, Mattias Villani and Robert Kohn	2010:245
The Output Gap, the Labor Wedge, and the Dynamic Behavior of Hours by Luca Sala, Ulf Söderström and Antonella Trigari	2010:246
Density-Conditional Forecasts in Dynamic Multivariate Models by Michael K. Andersson, Stefan Palmqvist and Daniel F. Waggoner	2010:247
Anticipated Alternative Policy-Rate Paths in Policy Simulations by Stefan Laséen and Lars E. O. Svensson	2010:248
MOSES: Model of Swedish Economic Studies by Gunnar Bårdsen, Ard den Reijer, Patrik Jonasson and Ragnar Nymoen	2011:249
The Effects of Endogenuos Firm Exit on Business Cycle Dynamics and Optimal Fiscal Policy by Lauri Vilmi	2011:250
Parameter Identification in a Estimated New Keynesian Open Economy Model by Malin Adolfson and Jesper Lindé	2011:251
Up for count? Central bank words and financial stress by Marianna Blix Grimaldi	2011:252
Wage Adjustment and Productivity Shocks by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	2011:253

Stylized (Arte) Facts on Sectoral Inflation by Ferre De Graeve and Karl Walentin	2011:254
Hedging Labor Income Risk by Sebastien Betermier, Thomas Jansson, Christine A. Parlour and Johan Walden	2011:255
Taking the Twists into Account: Predicting Firm Bankruptcy Risk with Splines of Financial Ratios by Paolo Giordani, Tor Jacobson, Erik von Schedvin and Mattias Villani	2011:256
Collateralization, Bank Loan Rates and Monitoring: Evidence from a Natural Experiment by Geraldo Cerqueiro, Steven Ongena and Kasper Roszbach	2012:257
On the Non-Exclusivity of Loan Contracts: An Empirical Investigation by Hans Degryse, Vasso Ioannidou and Erik von Schedvin	2012:258
Labor-Market Frictions and Optimal Inflation by Mikael Carlsson and Andreas Westermark	2012:259
Output Gaps and Robust Monetary Policy Rules by Roberto M. Billi	2012:260
The Information Content of Central Bank Minutes by Mikael Apel and Marianna Blix Grimaldi	2012:261
The Cost of Consumer Payments in Sweden	2012:262
by Björn Segendorf and Thomas Jansson	
Trade Credit and the Propagation of Corporate Failure: An Empirical Analysis	2012:263
by Tor Jacobson and Erik von Schedvin	
Structural and Cyclical Forces in the Labor Market During the Great Recession: Cross-Country Evidence	2012:264
by Luca Sala, Ulf Söderström and AntonellaTrigari	
Pension Wealth and Household Savings in Europe: Evidence from SHARELIFE	2013:265
by Rob Alessie, Viola Angelini and Peter van Santen	
Long-Term Relationship Bargaining	2013:266
by Andreas Westermark	
Using Financial Markets To Estimate the Macro Effects of Monetary Policy: An Impact-Identified FAVAR*	2013:267
by Stefan Pitschner	
DYNAMIC MIXTURE-OF-EXPERTS MODELS FOR LONGITUDINAL AND DISCRETE-TIME SURVIVAL DATA	2013:268
by Matias Quiroz and Mattias Villani	
Conditional euro area sovereign default risk	2013:269
by André Lucas, Bernd Schwaab and Xin Zhang	
Nominal GDP Targeting and the Zero Lower Bound: Should We Abandon Inflation Targeting?*	2013:270
by Roberto M. Billi	
Un-truncating VARs*	2013:271
by Ferre De Graeve and Andreas Westermark	
Housing Choices and Labor Income Risk	2013:272
by Thomas Jansson	
Identifying Fiscal Inflation*	2013:273
by Ferre De Graeve and Virginia Queijo von Heideken	0040.074
On the Redistributive Effects of Inflation: an International Perspective"	2013:274
by Paola Boel	0040.075
Business Cycle Implications of Mongage Spreads	2013:275
by Kall Walefull	2012:276
economic models by Isajah Hull	2013.270
A detrimental feedback loop: deleveraging and adverse selection	2013-277
by Christoph Bartsch	2013.211
Distortionary Fiscal Policy and Monetary Policy Goals	2013-278
by Klaus Adam and Roberto M Billi	2010.270
Predicting the Spread of Financial Innovations: An Enidemiological Approach	2013.279
by Isaiah Hull	
Firm-Level Evidence of Shifts in the Supply of Credit	2013:280

Lines of Credit and Investment: Firm-Level Evidence of Real Effects of the Financial Crisis by Karolina Holmberg	2013:281
A wake-up call: information contagion and strategic uncertainty by Toni Ahnert and Christoph Bertsch	2013:282
Debt Dynamics and Monetary Policy: A Note	2013:283
by Stefan Laséen and Ingvar Strid	
Optimal taxation with home production	2014:284
by Conny Olovsson	
Incompatible European Partners? Cultural Predispositions and Household Financial Behavior	2014:285
by Michael Haliassos, Thomas Jansson and Yigitcan Karabulut	
How Subprime Borrowers and Mortgage Brokers Shared the Piecial Behavior	2014:286
by Antje Berndt, Burton Hollifield and Patrik Sandås	
The Macro-Financial Implications of House Price-Indexed Mortgage Contracts	2014:287
by Isaiah Hull	
Does Trading Anonymously Enhance Liquidity?	2014:288
by Patrick J. Dennis and Patrik Sandås	
Systematic bailout guarantees and tacit coordination	2014:289
by Christoph Bertsch, Claudio Calcagno and Mark Le Quement	
Selection Effects in Producer-Price Setting	2014:290
by Mikael Carlsson	
Dynamic Demand Adjustment and Exchange Rate Volatility	2014:291
by Vesna Corbo	
Forward Guidance and Long Term Interest Rates: Inspecting the Mechanism	2014:292
by Ferre De Graeve, Pelin Ilbas & Raf Wouters	
Firm-Level Shocks and Labor Adjustments	2014:293
by Mikael Carlsson, Julián Messina and Oskar Nordström Skans	
A wake-up call theory of contagion	2015:294
by Toni Ahnert and Christoph Bertsch	
Risks in macroeconomic fundamentals and excess bond returns predictability	2015:295
by Rafael B. De Rezende	
The Importance of Reallocation for Productivity Growth: Evidence from European and US Banking	2015:296
by Jaap W.B. Bos and Peter C. van Santen	
SPEEDING UP MCMC BY EFFICIENT DATA SUBSAMPLING	2015:297
by Matias Quiroz, Mattias Villani and Robert Kohn	
Amortization Requirements and Household Indebtedness: An Application to Swedish-Style Mortgages	2015:298
by Isaiah Hull	
Fuel for Economic Growth?	2015:299
by Johan Gars and Conny Olovsson	
Searching for Information	2015:300
by Jungsuk Han and Francesco Sangiorgi	
What Broke First? Characterizing Sources of Structural Change Prior to the Great Recession	2015:301
by Isaiah Hull	
Price Level Targeting and Risk Management	2015:302
by Roberto Billi	
Central bank policy paths and market forward rates: A simple model	2015:303
by Ferre De Graeve and Jens Iversen	
Jump-Starting the Euro Area Recovery: Would a Rise in Core Fiscal Spending Help the Periphery?	2015:304
by Olivier Blanchard, Christopher J. Erceg and Jesper Lindé	
Bringing Financial Stability into Monetary Policy*	2015:305
by Eric M. Leeper and James M. Nason	
SCALABLE MCMC FOR LARGE DATA PROBLEMS USING DATA SUBSAMPLING AND THE DIFFERENCE ESTIMATOR	2015:306
by MATIAS QUIROZ, MATTIAS VILLANI AND ROBERT KOHN	

Modeling financial sector joint tail risk in the euro area 2015 by André Lucas Bernd Schwash and Xin Zhang	5:308
NY ANNIE LUUAS, DENNU SUNWAAN ANN ANN ZINANY	
Score Driven Exponentially Weighted Moving Averages and Value-at-Risk Forecasting 2015	5:309
by André Lucas and Xin Zhang	
On the Theoretical Efficacy of Quantitative Easing at the Zero Lower Bound 2015	5:310
by Paola Boel and Christopher J. Waller	
Optimal Inflation with Corporate Taxation and Financial Constraints 2015	5:311
by Daria Finocchiaro. Giovanni Lombardo. Caterina Mendicino and Philippe Weil	
Fire Sale Bank Recapitalizations 2015	5:312
by Christoph Bertsch and Mike Mariathasan	
Since you're so rich, you must be really smart: Talent and the Finance Wage Premium 2015	5:313
by Michael Böhm. Daniel Metzger and Per Strömberg	
Debt. equity and the equity price puzzle 2015	314
by Daria Finocchiaro and Caterina Mendicino	
Trade Credit: Contract-Level Evidence Contradicts Current Theories 2016	·315
by Tore Ellingsen, Tor, Jacobson and Erik von Schedvin	
Double Liability in a Branch Banking System: Historical Evidence from Canada 2016	
buble Elability in a Branch Banking System. Historical Evidence from Canada 2010	5.510
Subprime Perrowers, Securitization and the Transmission of Publicase Cucles 2016	
Subprime Borrowers, Securitization and the Transmission of Business Cycles 2010	0.317
Deal Time Forecasting for Manaton / Deliny Analysis: The Case of Quarinee Dilaborty 2010	
Real-Time Forecasting for Monetary Policy Analysis: The Case of Sveriges Riksbank 2016 by Jaco Jugram, Stafen Jacob Jugram, Ju	518
by Jens Iversen, Steran Laseen, Henrik Lundvall and Uli Soderstrom	
Fed Liftoff and Subprime Loan Interest Rates: Evidence from the Peer-to-Peer Lending 2016	5:319
by Christoph Bertsch, Isaian Hull and Xin Zhang	
Curbing Shocks to Corporate Liquidity: The Role of Trade Credit 2016	5:320
by Niklas Amberg, Tor Jacobson, Erik von Schedvin and Robert Townsend	
Firms' Strategic Choice of Loan Delinquencies 2016	5:321
by Paola Morales-Acevedo	
Fiscal Consolidation Under Imperfect Credibility 2016	5:322
by Matthieu Lemoine and Jesper Lindé	
Challenges for Central Banks' Macro Models 2016	5:323
by Jesper Lindé, Frank Smets and Rafael Wouters	
The interest rate effects of government bond purchases away from the lower bound 2016	6:324
by Rafael B. De Rezende	
COVENANT-LIGHT CONTRACTS AND CREDITOR COORDINATION 2016	325
by Bo Becker and Victoria Ivashina	
Endogenous Separations, Wage Rigidities and Employment Volatility 2016	6:326
by Mikael Carlsson and Andreas Westermark	
Renovatio Monetae: Gesell Taxes in Practice 2016	6:327
by Roger Svensson and Andreas Westermark	
Adjusting for Information Content when Comparing Forecast Performance 2016	5:328
by Michael K. Andersson, Ted Aranki and André Reslow	
Economic Scarcity and Consumers' Credit Choice 2016	329
by Marieke Bos, Chloé Le Coq and Peter van Santen	
Uncertain pension income and household saving 2016	6:330
by Peter van Santen	
Money, Credit and Banking and the Cost of Financial Activity 2016	5:331
by Paola Boel and Gabriele Camera	
Oil prices in a real-business-cycle model with precautionary demand for oil 2016	5:332
by Conny Olovsson	
Financial Literacy Externalities 2016	3:333
by Michael Haliasso, Thomas Jansson and Yigitcan Karabulut	

The timing of uncertainty shocks in a small open economy by Hanna Armelius, Isaiah Hull and Hanna Stenbacka Köhler	2016:334
Quantitative easing and the price-liquidity trade-off by Marien Ferdinandusse, Maximilian Freier and Annukka Ristiniemi	2017:335
What Broker Charges Reveal about Mortgage Credit Risk by Antie Berndt, Burton Hollifield and Patrik Sandåsi	2017:336
Asymmetric Macro-Financial Spillovers	2017:337
by Kristina Bluwstein	2011.001
Latency Arbitrage When Markets Become Faster	2017:338
by Burton Hollifield. Patrik Sandås and Andrew Todd	2011.000
How big is the toolbox of a central banker? Managing expectations with policy-rate forecasts: Evidence from Sweden	2017:339
by Magnus Åhl	
International business cycles: quantifying the effects of a world market for oil	2017:340
by Johan Gars and Conny Olovsson I	
Systemic Risk: A New Trade-Off for Monetary Policy?	2017:341
by Stefan Laséen, Andrea Pescatori and Jarkko Turunen	
Household Debt and Monetary Policy: Revealing the Cash-Flow Channel	2017:342
by Martin Flodén, Matilda Kilström, Jósef Sigurdsson and Roine Vestman	
House Prices, Home Equity, and Personal Debt Composition	2017:343
by Jieying Li and Xin Zhang	
Identification and Estimation issues in Exponential Smooth Transition Autoregressive Models	2017:344
by Daniel Buncic	
Domestic and External Sovereign Debt	2017:345
by Paola Di Casola and Spyridon Sichlimiris	
The Role of Trust in Online Lending by Christoph Bertsch, Isaiah Hull, Yingjie Qi and Xin Zhang	2017:346
On the effectiveness of loan-to-value regulation in a multiconstraint framework by Anna Grodecka	2017:347
Shock Propagation and Banking Structure by Mariassunta Giannetti and Farzad Saidi	2017:348
The Granular Origins of House Price Volatility	2017:349
by Isaiah Hull, Conny Olovsson, Karl Walentin and Andreas Westermark	
Should We Use Linearized Models To Calculate Fiscal Multipliers?	2017:350
by Jesper Lindé and Mathias Trabandt	
The impact of monetary policy on household borrowing – a high-frequency IV identification by Maria Sandström	2018:351
Conditional exchange rate pass-through: evidence from Sweden by Vesna Corbo and Paola Di Casola	2018:352
Learning on the Job and the Cost of Business Cycles by Karl Walentin and Andreas Westermark	2018:353
Trade Credit and Pricing: An Empirical Evaluation by Niklas Amberg, Tor Jacobson and Erik von Schedvin	2018:354
A shadow rate without a lower bound constraint by Rafael B. De Rezende and Annukka Ristiniemi	2018:355
Reduced "Border Effects", FTAs and International Trade by Sebastian Franco and Erik Frohm	2018:356
Spread the Word: International Spillovers from Central Bank Communication by Hanna Armelius, Christoph Bertsch, Isaiah Hull and Xin Zhang	2018:357
Predictors of Bank Distress: The 1907 Crisis in Sweden by Anna Grodecka, Seán Kenny and Anders Ögren	2018:358



Sveriges Riksbank Visiting address: Brunkebergs torg 11 Mail address: se-103 37 Stockholm

Website: www.riksbank.se Telephone: +46 8 787 00 00, Fax: +46 8 21 05 31 E-mail: registratorn@riksbank.se